Author’s response to reviews

Title: Influence of typical handball characteristics on upper body posture and postural control in male handball players

Authors:

Daniela Ohlendorf (ohlendorf@med.uni-frankfurt.de)
Saskia Salzer (salzer.saskia@gmx.de)
Raffaela Haensel (RaffaelaHaensel@web.de)
Julia Rey (rey@med.uni-frankfurt.de)
Laura Maltry (maltry@med.uni-frankfurt.de)
Fabian Holzgreve (holzgreve@med.uni-frankfurt.de)
Jasmin Lampe (j.lampe@med.uni-frankfurt.de)
Eileen Wanke (wanke@med.uni-frankfurt.de)
David Groneberg (Groneberg@med.uni-frankfurt.de)

Version: 3 Date: 02 Jan 2020

Author’s response to reviews:

1. Please provide a “List of abbreviations” subsection after the Conclusions section. If abbreviations are used in the text, they should be defined in the text at first use and included in this list.

◊ We added a “List of abbreviations” subsection after the Conclusions section

2. The Availability of data and materials section refers to the raw data used in your study and presenting tables and figures is not sufficient to state that all data is contained within the manuscript and additional files. Please only use this statement if you have indeed provided all raw data on which your study is based. We strongly encourage all authors to share their raw data, either by providing it in a supplementary file, making it available on request to the authors, or depositing it in a public repository and providing the details on how to access it in this section. If you do not wish to share your data, please clearly state this in this section along with a justification. Data availability statements can take one of the following forms (or a combination of more than one if required for multiple datasets):

◊ The datasets used and/or analysed during the current study are available from the corresponding author on reasonable request.
3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

◊ We did so.